EO-1 Weekly Status Week of Mar 24 – Mar 30, 2006

Day of Year 083-089
Mission Day 1949 - 1955
EO-1 General:

The EO-1 satellite performed nominally this week.  There were 140 Data Collection Events (DCEs) planned. 
EO-1 received augmented funding for FY 2006 for the Mission Science Office (MSO) and some augment operations funding to enable continued support of special science requests and ongoing science support.  In particular, the MSO is analyzing at what rate EO-1 can be lowered and still retain compatibility with Landsat7 to act as a Landsat7 data gap filler.

The USGS Earth Resources Observations and Science (EROS) Data Center facility has posted a notice on their EO-1 website (http://eo1.usgs.gov) that the mission has been extended through FY07 with the possibility for further extensions.  During this extended mission, as a fuel saving measure, the EO-1 orbit is being lowered in small increments that have taken EO-1 out of formation flying with Landsat7.  Information on EO-1 operations during this extended mission period and the USGS/EROS service to the EO-1 user community can be obtained from the above stated website.  
Spacecraft:

All spacecraft subsystems performed nominally this week.  

Instruments:

The instruments operated nominally this week. 

Technology Activities:
The EO-1 extended mission phase has been transformed into an on-orbit testbed for advanced technology and hyperspectral research.  The status of various validation efforts is as follows:

Sensor Web & Virtual Observatory Demonstrations

The automated schedule generation software continued to operate smoothly producing all weekly and daily activity schedules without problems.  All sensor web replacements (triggered scenes from other sensor installations) were uplinked automatically.  All science priority replacements were also uplinked automatically. 
Autonomous Sciencecraft Experiment (ASE)

ASE controlled the spacecraft all week.  JPL analysts continue to investigate the recent command delays that appear to be associated with high WARP CPU utilization.
Hybrid Ground Phased Array Antenna Validation (funded & in process) 

We retested the space fed lens aperture and it appears that the aperture efficiency is much lower than anticipated, i.e. 3%.  We expected an aperture efficiency of 50%.  We have scheduled a telecom with team members to see if we understand what is happening.  At 3% efficiency, we will not be able to make the adaptive algorithms converge to enable capture of the bits from SAC-C.  We will try to determine if this is a fabrication flaw or that this low aperture efficiency is inherent in space fed lens design in general.

Onboard and Ground-based Compression of Hyperspectral Data (unfunded & in definition)

Nothing new to report.

Precision Attitude Control Using the Three Axis Magnetometer (TAM) (unfunded and in definition) 
Nothing new to report.

Livingstone Onboard Model Based Diagnostic Tools 


Nothing new to report.


Autonomy Testbed
CHIPS funding expired on March 30, 2006.  At this time we have not heard whether funding will be provided for continued operations.  We will continue to prepare the next step to moving mobile agents onboard CHIPS in case CHIPS operations continue.

Ground System

Hardware Status

The EO-1 internal network switch (inside the firewall) was replaced on 03/29/2006 (06-088) due to a noisy fan.  All EO-1 MOC systems are currently using the new switch with the exception of DPS03 (S/C time sorted telemetry, EDOS science data incoming deliveries, Level-0 outgoing deliveries).  It appears that the network card in DPS03 and the new switch are not auto-negotiating properly.  Options are to change the BIOS switches for the network card in dps03 or to daisy chain a small switch out of the new switch for DPS03.  Until that time, the old switch is being daisy chained off the new switch to keep DPS03 connected. 

Ground and Space EOS Data and Operations System (EDOS), GSFC, MD, USA
There are no major problems to report.

DataLynx Ground Station (PF1), Poker Flat, AK, USA

There are no major problems to report.

Alaska Ground Station (AGS), Poker Flat, AK, USA 

AGS is currently “Red” due to damaged cables.  EO-1 passes with AGS have either been moved to PF1 or deleted.

Svalbard Ground Station (SGS), Longyearbyen, Norway

There are no major problems to report.

Tracking and Data Relay Satellite System (TDRSS), White Sands, NM, USA


There are no major problems to report.


USGS Multi-Satellite Ground Station (LGS), Sioux Falls, SD, USA


LGS is now being scheduled only for contingencies.


Tasmanian Earth Resources Satellite Station, a.k.a. Hobart (HGS), Hobart, Australia

There are no major problems to report.  

Cordoba Ground Station (CGS), Cordoba, Argentina


No passes were scheduled this week. 

Upcoming Events:

· MLT control maneuver on 04/04/2006 (06-094)

· PPT operations on (or about) 04/18/2006 (06-108)
Imagery Status:

Scenes and Engineering Cals planned for week of Mar 24 – Mar 30, 2006                 140
Total scenes and engineering calibrations planned for entire mission – approximately 27,237
Total Scenes:  ALI scenes in the Level 0 archive            25,669 (as of Mar 30, 2006)

                        Hyperion scenes in the Level 0 archive    25,431
Publications and Presentations Status (as of 03-09-06):

341 publications 

234 external presentations

  49 articles and press releases
















Blocker Plate




















































































































































































































































































































































































































































































































































































Page 3 of 3

